# Temporary treatment cessation compared with continuation of tyrosine kinase inhibitors for adults with renal cancer: the STAR non-inferiority RCT

Fiona Collinson,<sup>1</sup> Kara-Louise Royle,<sup>1</sup> Jayne Swain,<sup>1</sup> Christy Ralph,<sup>2</sup> Anthony Maraveyas,<sup>3</sup> Tim Eisen,<sup>4</sup> Paul Nathan,<sup>5</sup> Robert Jones,<sup>6</sup> David Meads,<sup>7</sup> Tze Min Wah,<sup>8</sup> Adam Martin,<sup>7</sup> Janine Bestall,<sup>9</sup> Christian Kelly-Morland,<sup>10</sup> Christopher Linsley,<sup>1</sup> Jamie Oughton,<sup>1</sup> Kevin Chan,<sup>11</sup> Elisavet Theodoulou,<sup>12</sup> Gustavo Arias-Pinilla,<sup>12</sup> Amy Kwan,<sup>13</sup> Luis Daverede,<sup>14</sup> Catherine Handforth,<sup>12</sup> Sebastian Trainor,<sup>15</sup> Abdulazeez Salawu,<sup>13</sup> Christopher McCabe,<sup>16</sup> Vicky Goh,<sup>17</sup> David Buckley,<sup>18</sup> Jenny Hewison,<sup>9</sup> Walter Gregory,<sup>1</sup> Peter Selby,<sup>2</sup> Julia Brown<sup>1</sup> and Janet Brown<sup>12\*</sup> on behalf of all the STAR investigators

<sup>&</sup>lt;sup>1</sup>Leeds Institute of Clinical Trials Research, University of Leeds, Leeds, UK <sup>2</sup>Leeds Institute of Medical Research, St James's University Hospital, University of Leeds, Leeds, UK

<sup>&</sup>lt;sup>3</sup>Academic Oncology, Faculty of Health Sciences, Hull York Medical School, Queens Centre Oncology and Haematology, Hull, UK

<sup>&</sup>lt;sup>4</sup>Department of Oncology, University of Cambridge and Cambridge University Hospitals NHS Foundation Trust, Addenbrooke's Hospital, Cambridge, UK <sup>5</sup>Department of Oncology, Mount Vernon Cancer Centre, East and North Hertfordshire NHS Trust, Hertfordshire, UK

<sup>&</sup>lt;sup>6</sup>School of Cancer Sciences, University of Glasgow, Beatson West of Scotland Cancer Centre, Glasgow, UK

<sup>&</sup>lt;sup>7</sup>Academic Unit of Health Economics, University of Leeds, Leeds, UK

<sup>&</sup>lt;sup>8</sup>Department of Diagnostic and Interventional Radiology, Leeds Teaching Hospitals Trust, Leeds, UK

<sup>&</sup>lt;sup>9</sup>Leeds Institute of Health Sciences, University of Leeds, Leeds, UK

<sup>&</sup>lt;sup>10</sup>Department of Cancer Imaging, King's College London, London, UK

<sup>&</sup>lt;sup>11</sup>Medical Oncology, Weston Park Cancer Hospital, Sheffield, UK

<sup>&</sup>lt;sup>12</sup>Division of Clinical Medicine, University of Sheffield, Weston Park Hospital, Sheffield, UK

<sup>&</sup>lt;sup>13</sup>Academic Unit of Clinical Oncology, University of Sheffield, Sheffield, UK

<sup>&</sup>lt;sup>14</sup>Department of Clinical Oncology, Austral University Hospital, Buenos Aires, Argentina

Published August 2024 DOI: 10.3310/JWTR4127

# Plain language summary

Temporary treatment cessation compared with continuation of tyrosine kinase inhibitors for adults with renal cancer: the STAR non-inferiority RCT

Health Technology Assessment 2024; Vol. 28: No. 45

DOI: 10.3310/JWTR4127

NIHR Journals Library www.journalslibrary.nihr.ac.uk

<sup>&</sup>lt;sup>15</sup>St James's Institute of Oncology, St James's University Hospital, Leeds, UK

<sup>&</sup>lt;sup>16</sup>Institute of Health Economics, University of Alberta, Edmonton, Canada

<sup>&</sup>lt;sup>17</sup>School of Biomedical Engineering and Imaging Sciences, King's College London, London, UK

<sup>&</sup>lt;sup>18</sup>Faculty of Medicine and Health, School of Medicine, University of Leeds, Leeds, UK

<sup>\*</sup>Corresponding author j.e.brown@sheffield.ac.uk

# Plain language summary

reatment breaks in cancer are of significant interest to patients and health professionals.

Renal cell carcinoma is the most common type of kidney cancer. Sunitinib and pazopanib are both targeted treatments. They were commonly used to treat advanced kidney cancer but often cause side effects, sometimes requiring use of a reduced dose or even stopping treatment.

The STAR trial was designed to see whether planned treatment breaks made patients with advanced kidney cancer being treated with sunitinib and pazopanib feel better, without substantially affecting how well the treatment worked. After 24 weeks of treatment, patients took sunitinib and pazopanib either as they normally would or in the alternative way with planned treatment breaks. Treating patients in this way was continued until drug-related side effects stopped treatment, patients' disease worsened while taking treatment or the patient died. The trial compared how well the different treatment strategies worked in terms of how long patients lived and their quality of life over that time.

This trial is the largest United Kingdom trial in advanced renal cell carcinoma. Patients took part from 60 United Kingdom centres between 2012 and 2017. It was funded by the National Institute for Health and Care Research Health Technology Assessment Programme and run by the Leeds Clinical Trials Research Unit.

In total, 920 patients took part. Four hundred and sixty-one patients were allocated to continue treatment and 459 were allocated to start at least one treatment break. Treatment breaks lasted on average 87 days. The length of time patients lived in both arms of the trial appeared similar, but this cannot be concluded due to insufficient information. Being allocated to have treatment breaks rather than continuing treatment did not negatively impact a patient's quality of life. Additionally, allocating patients to have treatment breaks was shown to have significant cost savings compared to just continuing treatment. Importantly planned treatment breaks were shown to be feasible.

# **Health Technology Assessment**

ISSN 2046-4924 (Online)

Impact factor: 3.6

A list of Journals Library editors can be found on the NIHR Journals Library website

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2022 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), EMBASE (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

### Criteria for inclusion in the Health Technology Assessment journal

Manuscripts are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

## **HTA** programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

### This article

The research reported in this issue of the journal was funded by the HTA programme as award number 09/91/21. The contractual start date was in June 2011. The draft manuscript began editorial review in December 2021 and was accepted for publication in March 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' manuscript and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this article.

This article presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

This article was published based on current knowledge at the time and date of publication. NIHR is committed to being inclusive and will continually monitor best practice and guidance in relation to terminology and language to ensure that we remain relevant to our stakeholders.

Copyright © 2024 Collinson *et al.* This work was produced by Collinson *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).